Author's response to reviews

Title: Temporal trends in female breast cancer mortality in Brazil and correlations with social inequalities: Ecological time-series study.

Authors:

Carolina MR Gonzaga (cmrgonzaga@gmail.com)
Ruffo Freitas-Junior PhD (ruffojr@terra.com.br)
Maria-Paula Curado PhD (maria-paula.curado@i-pri.org)
Ana-Luiza L Sousa PhD (demmillima@gmail.com)
José-Augusto Souza-Neto (augustonetojs@hotmail.com)
Marta R Souza PhD (martary@gmail.com)

Version: 2
Date: 17 June 2014

Author's response to reviews: see over
BMC Public Health

Research article: Temporal trends in female breast cancer mortality in Brazil and correlations with social inequalities: Ecological time-series study.

Carolina MR Gonzaga, Ruffo Freitas-Junior PhD, Maria-Paula Curado PhD, Ana-Luiza L Sousa PhD, José-Augusto Souza-Neto and Marta R Souza PhD

Goiania, June 16th, 2014.

Dear Reviewers,

We prepared this letter in order to respond each of your considerations.

(1) Abstract: we added the background and minimized the use of abbreviations;

(2) Manuscript Body: We changed the ‘Introduction’ section heading to ‘Background’;

(3) We included:

- A List of abbreviation used;
- Competing interests and financial section between Conclusion and Author’s contributions.
- Authors' contributions

Sincerely,

Prof. Ruffo Freitas-Junior, MD, PhD.